Author’s response to reviews

Title: How Shoulder Immobilization Influences Daily Physical Activity – An Accelerometer Based Preliminary Study

Authors:

Carolin Rickert (carolin.rickert@ukmuenster.de)
Monika Grabowski (m_grab06@uni-muenster.de)
Georg Gosheger (georg.gosheger@ukmuenster.de)
Dominik Schorn (dominik.schorn@ukmuenster.de)
Kristian Nikolaus Schneider (kristian.schneide@ukmuenster.de)
Sebastian Klingebiel (sebastian.klingebiel@ukmuenster.de)
Dennis Liem (liem@sporthopaedicum.de)

Version: 1 Date: 15 Oct 2019

Author’s response to reviews:

Dear Sir or Madame,

thank you for your comments.

The revised manuscript is uploaded.

Thank you for your consideration,

sincerely,

Carolin Rickert